Prolonged multi-phase magmatism due to plume lithosphere interaction as applied to the High Arctic Large Igneous Province

Björn H. Heyn¹, Grace Shephard², and Clinton P. Conrad³

¹Centre for Planetary Habitability ²University of Oslo ³Centre for Earth Evolution and Dynamics

December 1, 2023

Abstract

The widespread High Arctic Large Igneous Province (HALIP) exhibits prolonged melting over more than 50 Myr, an observation that is difficult to reconcile with the classic view of large igneous provinces and associated melting in plume heads. Hence, the suggested plume-related origin and classification of HALIP as a large igneous province have been questioned. Here, we use numerical models that include melting and melt migration to investigate a rising plume interacting with variable lithosphere thickness, i.e. an extended-basin-to-craton setting. Models reveal significant spatial and temporal variations in melt volumes and pulses of melt production, including protracted melting for at least about 30-40 Myr, but only if migrating melt transports heat upwards and enhances local lithospheric thinning. Plume material deflected from underneath the Greenland craton can then reactivate melting zones below the previously plume-influenced Sverdrup Basin, even though the plume is already ~500 km away. Hence, melting zones may not represent the location of the deeper plume stem at a given time. Plume flux pulses associated with mantle processes or magma processes within the crust may alter the timing and volume of secondary pulses and their surface expression. Our models suggest that HALIP magmatism is expected to exhibit plume-related trace element signatures throughout time, but potentially shift from mostly tholeiitic magmas in the first pulse towards more alkalic compositions for secondary pulses, with regional variations in timing of magma types. We propose that the prolonged period of rejuvenated magmatism of HALIP is consistent with plume impingement on a cratonic edge.

Prolonged multi-phase magmatism due to plume lithosphere interaction as applied to the High Arctic Large Igneous Province

Björn H. Heyn¹, Grace E. Shephard^{1,2}, Clinton P. Conrad¹

 $^1{\rm Centre}$ for Planetary Habitability, Department of Geosciences, University of Oslo, Norway $^2{\rm Research}$ School of Earth Sciences, Australian National University, Australia

Key Points:

1

2

3

4

5

7

8	• Mantle plumes interacting with changes in lithosphere thickness at craton edges
9	can cause prolonged melting with pulses in the same region
10	• Rejuvenated melting happens underneath previously melt-affected thinned litho-
11	sphere several hundred km downstream of the plume stem
12	• The timing and duration of rejuvenated melting in models correspond to and there
13	fore may explain observations of magmatic pulses from HALIP

Corresponding author: Björn H. Heyn, b.h.heyn@geo.uio.no

14 Abstract

The widespread High Arctic Large Igneous Province (HALIP) exhibits prolonged melt-15 ing over more than 50 Myr, an observation that is difficult to reconcile with the classic 16 view of large igneous provinces and associated melting in plume heads. Hence, the sug-17 gested plume-related origin and classification of HALIP as a large igneous province have 18 been questioned. Here, we use numerical models that include melting and melt migra-19 tion to investigate a rising plume interacting with variable lithosphere thickness, i.e. an 20 extended-basin-to-craton setting. Models reveal significant spatial and temporal vari-21 ations in melt volumes and pulses of melt production, including protracted melting for 22 at least about 30-40 Myr, but only if migrating melt transports heat upwards and en-23 hances local lithospheric thinning. Plume material deflected from underneath the Green-24 land craton can then re-activate melting zones below the previously plume-influenced 25 Sverdrup Basin, even though the plume is already $\sim 500 \,\mathrm{km}$ away. Hence, melting zones 26 may not represent the location of the deeper plume stem at a given time. Plume flux pulses 27 associated with mantle processes or magma processes within the crust may alter the tim-28 ing and volume of secondary pulses and their surface expression. Our models suggest that 29 HALIP magmatism is expected to exhibit plume-related trace element signatures through-30 out time, but potentially shift from mostly tholeiitic magmas in the first pulse towards 31 more alkalic compositions for secondary pulses, with regional variations in timing of magma 32 types. We propose that the prolonged period of rejuvenated magnatism of HALIP is con-33 sistent with plume impingement on a cratonic edge. 34

35 Plain Language Summary

Typically, large mantle upwellings ("mantle plumes") are expected to cause catas-36 trophic large-scale but short-lived (within a few million years) volcanism. However, a mas-37 sive past volcanic event now distributed onshore and offshore across the Arctic (the High 38 Arctic Large Igneous Province - HALIP) defies this expectation. This wide-spread mag-39 matism exhibits dates spanning more than 50 Myrs, with several pulses of increased ac-40 tivity. Based on this prolonged magmatism, it has been questioned whether all of it can 41 be attributed to a mantle plume, despite the geochemistry of basalts indicating a plume 42 source. Here, we show that a plume can cause prolonged melting including pulses of mag-43 matism if it interacts with an increase in lithosphere thickness. Once the plume moved 44 below the thicker lithosphere, hot plume material is channeled along the base of the litho-45 sphere towards the adjacent thinner part, where it can reactivate previous melting re-46 gions. At this time, the active plume can be about 500 km away from the melting region, 47 hence plume-related melt cannot be used as a proxy for the plume position at the given 48 time. Based on the models, we suggest that the prolonged HALIP magmatism was caused 49 by a plume impinging on the edge of a craton. 50

51 **1** Introduction

Located at the geographic top of the world, the High Arctic Large Igneous Province 52 (HALIP; (Tarduno, 1998; Maher, 2001)) is one of the most enigmatic volcanic provinces 53 on Earth. Broadly speaking, HALIP is attributed to a range of Cretaceous aged (~ 130 -54 80 Ma) volcanic and magmatic rocks, currently distributed onshore and offshore around 55 the circum-Arctic. HALIP includes flood basalts (both continental and oceanic), plu-56 tonic complexes, dykes, sills, and pyroclastic flows. They are dominantly tholeiitic in na-57 ture, but with numerous alkaline suites (e.g., Estrada et al., 2016). From roughly east 58 to west, localities include the Canadian Arctic Islands, northern Greenland, the central 59 Arctic Ocean (Alpha-Mendeleev Ridge), Svalbard, the Barents Shelf, Franz Josef Land, 60 the De Long Islands, and Siberian shelves (Figure 1). The large geographic footprint of 61 both intrusive and extrusive rocks is partly attributed to the mechanism of emplacement 62 (i.e. mantle plume arrival) as well as subsequent dispersal via post-emplacement tectonic 63

- ⁶⁴ motions (i.e. opening of the Eurasia Basin and Eurekan deformation). HALIP is also linked
- to significant regional oceanographic and climatic environmental changes (e.g., Galloway
- 66 et al., 2022).



Figure 1: Overview of the Arctic domain at present-day (bathymetry - IBCAO; Jakobsson et al., 2012) with HALIP localities in black lines (mapped or inferred dyke swarms), and orange and grey polygons (proposed HALIP regions beyond those of dykes) as compiled from various sources (Jowitt et al., 2014; Polteau et al., 2016; Døssing et al., 2017; Minakov et al., 2018); HAMD - High Arctic Magnetic Domain of Oakey and Saltus (2016). The short dashed orange contour is the -2500 m bathymetry line outlining the Alpha-Mendeleev Ridge; the star is the potential arrival site of the HALIP plume; the long dashed white line approximately marks the NW-SE transect traversing from the Amerasia Basin to Sverdrup Basin to the Greenland Craton as used in the numerical models (basin-margin-craton). AHI - Axel Heiberg Island. Note: at the initial HALIP pulse at around 122 Ma (Figure 2), the tectonic configuration was different - the Amerasia Basin was starting to open, the Eurasia Basin did not exist, and Eurekan deformation had not yet occurred.

As with many large-igneous provinces (LIPs) worldwide, a mantle plume is considered to be the primary source of HALIP volcanism. This is supported by several lines of evidence including the widespread distribution of HALIP rocks and the large volume of magmatism (both pointing to a large anomaly of elevated mantle temperatures e.g. (Coffin & Eldholm, 1994; Buchan & Ernst, 2018)), geochemistry containing a primitive mantle and/or recycled oceanic lithosphere component (pointing to a deep mantle source e.g. (Estrada, 2015); though alternative, shallow signals are discussed below), a pattern of radiating and circumferential dykes (pointing to a sub-circular mantle plume head im-

67

68

69

70

71

72

73

74

pinging on the lithosphere e.g., Buchan & Ernst, 2018; Minakov et al., 2018), and ma-75 jor regional sedimentary pathway reorganizations in the Barents Sea (pointing to sig-76 nificant relative uplift to the north e.g. (Midtkandal et al., 2020)). The arrival location 77 of the HALIP plume at approximately 130 Ma (Figure 2), is often reconstructed offshore, 78 in the region of the Alpha Ridge (e.g., Buchan & Ernst, 2018) or Lomonosov Ridge (Jackson 79 et al., 2010). Unlike many other LIPs, HALIP does not have a clear hotspot track, though 80 it may be linked to later Iceland plume dynamics and a potential trajectory beneath Green-81 land (e.g., Lawver & Müller, 1994). It is also worth noting that these observations are 82 complicated by a complex tectonic setting, with the the near-contemporaneous opening 83 (including episodes of rifting, hyperextension and/or seafloor-spreading) of the Amera-84 sia Basin to the north, as well as the later Eurekan deformation event (or orogeny) which 85 affected much of the eastern Canadian Arctic Islands, Northern Greenland, and Sval-86 bard (e.g. (Pease et al., 2014; Shephard et al., 2013)). 87

Alternatives to a mantle plume origin of HALIP-magmatism have been proposed, 88 at least to account for the later (post \sim 120 Ma e.g., Dockman et al., 2018) phases of HALIP 89 (second and third pulse at about 95 Ma and 81 Ma, Figure 2) and therefore its protracted 90 nature as a whole. Edge driven convection (as related to the inherted lithospheric struc-91 ture of region) is the dominant process invoked. Rifting of the Amerasia Basin (e.g., Teg-92 ner et al., 2011) or even Labrador and Baffin Bay rifting (e.g., Thorarinsson et al., 2011) 93 with associated decompression melting are also sometimes invoked as possible mecha-94 nisms for some of the HALIP magmatism. For the Canadian Arctic Islands region, Dockman 95 et al. (2018) questioned the need for a plume for the younger phases of HALIP, calling 96 upon multi-phase (decompression) melting and thermal erosion related to edge-driven 97 convection with or without shear, although they do not rule out that a plume may have influenced the asthenospheric flow. However, general modelling studies such as by Manjón-99 Cabeza Córdoba and Ballmer (2021, 2022) show that edge-driven convection can only 100 rarely produce and sustain magmatism, and generated volumes of magma are small. 101

Thus whether the HALIP rocks and their origins should be sub-divided temporally, 102 geographically or by mechanism remains an active question discussed within the Arc-103 tic geoscience community. Here we return to a plume hypothesis, and propose that its 104 arrival drives HALIP magmatism as a whole. A key question is therefore, can plume-105 lithosphere interactions explain the long-lived volcanism and pulses observed for HALIP? 106 A regionally-focused study is therefore pertinent; focusing on the HALIP localities in the 107 Canadian Arctic Island therefore offers an opportunity to evaluate the contributions from 108 both shallow and deeper mantle sources, as well as the temporal characteristics of melt-109 ing. Furthermore, numerical models of mantle convection, which incorporate plume and 110 lithospheric dynamics have never been applied to the High Arctic LIP. In this paper, we 111 suggest that a plume interacting with pre-existing lithosphere-asthenosphere boundary 112 (LAB) topography across the Arctic can produce prolonged and pulsating magmatism 113 across a large area, while other mechanisms may play a secondary role in the emplace-114 ment of HALIP. 115

¹¹⁶ 2 Constraints on the timing and geochemistry of HALIP magmatism

Documented ages of HALIP magmatism spread over a temporal range of ca. 50+ Myrs, 117 from $\sim 131 - 85$ Ma, with a potential peak phase around 122 Ma. If Kap Washington 118 volcanics of northern Greenland (71 - 61 Ma Estrada et al., 2010; Tegner et al., 2011;119 Thorarinsson et al., 2011) are also included, then this time frame increases to over 70 Myrs. 120 The protracted and/or multiple pulses of HALIP pose a particular challenge because most 121 traditionally-defined LIPs are erupted in a relatively short amount of time (1-5 Myr)122 for > 75% of the volume). Could a mantle plume potentially cause over 50 Myr of in-123 termittent and spatially variable melting? A particular element to this complexity is in 124 the geochemical signature of HALIP rocks, which show significant variations in compo-125 sition between different localities, including those within and between the Canadian and 126



Figure 2: Summary of HALIP dates and the three proposed pulses, coloured by Arctic location and divided into tholeiitic (circle) and alkaline (square) samples with 2 sigma error. Redrawn from the compilation of Dockman et al. (2018), which was in turn based on U-Pb and 40Ar/39Ar studies of Corfu et al. (2013); Estrada (2015); Estrada and Henjes-Kunst (2004); Evenchick et al. (2015); Jokat et al. (2013); Kontak et al. (2001); Villeneuve and Williamson (2006).

Eurasian sectors, and similar sample ages. Broadly speaking, there seems to be consen-127 sus that an upwelling mantle plume was involved (albeit the more ambiguous terms "as-128 thenospheric upwellings" or "decompressional melting" are sometimes referred to). How-129 ever, there is significant geochemical heterogeneity pointing to one, more or all of the 130 following contributions: a heterogeneous mantle plume (e.g. ocean island basalt or re-131 cycled oceanic crust signals), crustal contamination, entrainment of sub-continental lith-132 sopheric mantle (SCLM), and/or an enriched sub-lithospheric fossil subduction zone from 133 a nearby subducted slab (metasomatic signal e.g., Shephard et al., 2016; Hadlari et al., 134 2018).135

The arrival of a mantle plume is typically associated with (continental or oceanic) 136 flood basalts and tholeiitic suites, with alkaline magmatism frequently pre-, syn- or post-137 dating this. Alkaline magmatism in continental settings may form distally to the main 138 zone of extension (or the mantle plume) and can often be used to test whether the same 139 mantle is sampled between tholeiitic and alkaline suites. The later alkaline suites of HALIP 140 appear to be more low-degree mantle melts (Bédard et al., 2021) and/or deeper sourced 141 (Dockman et al., 2018) as compared to their tholeiitic counterparts. The alkaline sites 142 are also regionally confined towards the east (not found in Axel Heiberg Island), with 143

the extrusive alkaline lavas only found in the northern part of Ellesmere Island (e.g., Trettin, n.d.; Dockman et al., 2018) It is important to further summarize the geochemical
heterogeneity here in the context of underlying melt-inducing processes and tectono-magmatic
origins, and therefore the definition of HALIP.

The Sverdrup Basin is a ca. 1000 km along strike Carboniferous to Paleogene rift 148 basin at the northern edge of the North American continent, and encompasses Ellesmere, 149 Axel Heiberg and Melville islands, amongst others (collectively the Canadian Arctic Is-150 lands). The basin includes changes in crutal (Figure 3a) and lithospheric thickness (Fig-151 152 ure 3b) from north to south that reflect its long-lived tectonic history, and likely existed in the past as well. Recent field campaigns also revealed varied geochemical and isotopic 153 signals from HALIP. While dating methods vary, Dockman et al. (2018) summarized three 154 pulses of Canadian HALIP magmatism; 124 - 120 Ma, 99 - 91 Ma, and 85 - 77 Ma. As 155 in other HALIP localities, they are dominated by tholeiitic-type rocks, such as the Isach-156 sen Formation. These are proposed to have an enriched mantle (EM)-like signature and 157 widespread crustal contamination (e.g., Naber et al., 2020; Bédard, Saumur, et al., 2021). 158 However, there are also least two younger groups of alkaline rocks in the Sverdrup Basin 159 region, which are much smaller in volumes. Dockman et al. (2018) describe an overlap 160 period of tholeiitic and alkaline magmatism from ca.100-85 Ma. As recently detailed in 161 Bédard, Saumur, et al. (2021); Bédard et al. (2021), these alkaline suites include the ~ 96 Ma 162 Fulmar Suite (including Strand Fjord formation), which are suggested to resemble EM-163 type ocean island basalts with little crustal contamination and a widespread source (apatite-164 rich) similar to the dominant tholeiites. That said, however, the Fulmar Suite includes 165 Hassel Formation rocks which exhibit a depleted lower crust signal. Two additional, geochemically-166 similar alkaline suites include the 92-93 Ma Wootton Intrusive Complex with plutonic 167 rocks and the 83–73 Ma Audhild Bay Suite (also referred to as the Hansen Point vol-168 canics) with mafic alkaline rocks. These younger localities are thought to resemble HIMU 169 ocean island basalts but with some crustally derived signals, including potentially shal-170 low melting due to flat heavy rare earth profiles. Bédard et al. (2021) concluded that north-171 ern Ellesmere Island magmas are derived from variably sampled heterogeneous regions 172 in the sub-continental lithospheric mantle, and do not rule out a mantle plume source. 173

This geochemical diversity emphasizes the unique characteristics and challenging nature of disentangling shallow-and-deep HALIP processes. Perhaps unsurprisingly, an ocean basin to rifted margin to cratonic margin setting, which has undergone several phases of tectonic events, has a significant amount of structural and compositional variation in its crust and lithosphere (Figure 3). This heterogeneity may in turn have been reactivated or sampled during the more complicated processes of magma migration and fractionation, especially during successive melting episodes.

3 Methods

In order to investigate the dynamics of plume-lithosphere interactions of HALIP, 182 we run 2-D numerical models of mantle convection in Cartesian geometry. We focus on 183 modelling the presence of melt and test the impact of variable lithosphere thickness on 184 melt generation relevant to the emplacement of HALIP. Modelling is done using the open 185 source finite element code ASPECT v2.4.0 (Kronbichler et al., 2012; Dannberg & Heis-186 ter, 2016; Heister et al., 2017; Bangerth et al., 2022), which includes melting/freezing 187 and melt migration under Darcy's law, and heat advection by melt (Dannberg & Heis-188 ter, 2016; Dannberg et al., 2019). Since we focus on upper mantle dynamics, the effects 189 of compressibility and depth-dependence of parameters except for viscosity are small (Albers 190 & Christensen, 1996), hence we can use the Boussinesq approximation of incompress-191 ibility. We build on earlier 2-D models described in (Heyn & Conrad, 2022) with addi-192 tional initial conditions and further complexities of melting / freezing and melt migra-193 tion, as described below. 194



Figure 3: Geophysical datasets for the Canadian Arctic Island and Greenland region overlain with 1500 km transect from Figure 1 (white dashed line). a) Crustal thickness from ArcCRUST (Lebedeva-Ivanova et al., 2019), b) elastic thickness from (Steffen et al., 2018), and c) P-wave seismic velocity anomaly at 150 km depth from SL2013sv (Schaeffer & Lebedev, 2013)

To capture the proposed arrival location of the HALIP plume, we have chosen an 195 approximately NW-SE transect (present-day coordinates) which traverses the (incipi-196 ent) Amerasia Basin, the Sverdrup Basin of the Canadian Arctic Islands, the cratonic 197 Canadian shield and the northernmost Greenland craton. Two sets of domain sizes are 198 run, and were chosen to minimize boundary effects, while keeping domains small enough 199 to limit computational costs, as well as allow for a systematic study of different bound-200 ary conditions and plate configurations. In the horizontal direction, models have extents 201 of either 1500 km or 4000 km, with a vertical dimension of 600 km or 800 km, respectively. 202 The larger domains are necessary for models with a moving plate. We use adaptive mesh 203 refinement to resolve melt migration, with resolution varying between about 25 km in 204 areas of the upper mantle away from the plume, to about 3 km along the surface, the lithosphere-205 asthenosphere boundary (LAB) and within the mantle plume. Regions with active melt-206 ing can have higher resolution, depending on melt fractions and melt velocity. 207

Since the tectonic history of the Arctic and the relative motion of plates and the 208 mantle plume are not well constrained, a useful first approximation for HALIP might 209 be a model with a mantle plume arriving under a stagnant tectonic plate (no-slip con-210 ditions). This simulates a scenario in which there is no relative motion between the plume 211 and the plate, but does not necessarily imply that there was no plate motion at all. How-212 ever, the Arctic was actively undergoing tectonic motions including spreading and ex-213 tension during HALIP emplacement. Furthermore, it has been suggested that the plume 214 later passed underneath Greenland (e.g., Steinberger et al., 2019; Martos et al., 2018), 215 so it seems likely that there has been a relative motion between the plate and the plume. 216 Therefore, we investigate models with 2 different scenarios: 217



22. cases with imposed plate velocity: model domain with dimensions of 4000x800 km,
 22. up to two steps in lithosphere thickness moving over the plume, an imposed plate
 223 velocity of 2 cm/yr at the top, imposed plate velocities at the side walls to bal 224 ance inflow and outflow, and free-slip at the bottom

For both scenarios, the temperature is fixed along the top and bottom boundary to $273 \,\mathrm{K}$ 225 and 1623 K, respectively, and the initial temperature is described by a linear gradient 226 within the lithosphere of thickness d (defined by the 1400 K isotherm), and a linear gra-227 dient from the base of the lithosphere to the bottom of the domain. Present-day esti-228 mates of lithospheric thickness (Figure 3b) and upper mantle structure (Figure 3c), con-229 strained using seismic and gravity data, vary significantly across our transect. Due to 230 the long-lived and multiple-phases of regional tectonic deformation, it is likely that sim-231 ilar lithospheric heterogeneity existed also in the past. To account for this, the lithosphere 232 thickness d within the models at the starting condition varies between at least $50 \,\mathrm{km}$ in 233 the extended basin north of Greenland and up to 200 km for the Greenland craton. The 234 transition between the basin and the craton is simulated by either a gradual increase or 235 up to two 'steps' in the depth of the LAB. Details of the model setup are shown in the 236 next section when the respective model results are discussed. 237

In order to simulate a (thermal) mantle plume, a plume seed is added to the temperature field as a Gaussian-shaped anomaly of excess temperature 250 K, both as temperature boundary condition kept during the run, and as part of the initial condition within the lowest 50 km of the domain. Models are run going forward in time for 150 Myr, which should be sufficient to capture the HALIP melting dynamics. For models without imposed plate velocity, the plume seed is removed after 75 Myr in order to limit potential melting times to a duration relatable to HALIP (see e.g. Figure 2).

Viscosity is known to be a key parameter for mantle convection, and it has been shown to play a major role in plume-lithosphere interaction and the amount of lithosphere thinning associated with the plume (Heyn & Conrad, 2022). Since the presence of melt will reduce the overall viscosity of the respective region, viscosity is implemented into the model as a modified diffusion-dislocation creep that includes the effect of melt following Dannberg and Heister (2016), and a step below the asthenosphere:

$$\eta_{\rm eff} = \frac{\eta_j}{\eta_{\rm ref}} \left(\frac{1}{\eta_{\rm eff}^{\rm diff}} + \frac{1}{\eta_{\rm eff}^{\rm disl}} \right)^{-1} \cdot \exp\left(\alpha_\phi \phi\right) \cdot \exp\left(\alpha_\psi \psi\right) \tag{1}$$

252 with

251

253

269

$$\eta_{\text{eff}}^{i} = \frac{1}{2} A^{-\frac{1}{n_{i}}} d^{\frac{m_{i}}{n_{i}}} \dot{\epsilon}_{i}^{\frac{1-n_{i}}{n_{i}}} \exp\left(\frac{E_{i} + PV_{i}}{n_{i}RT}\right).$$
(2)

Parameters in equation (2) are the prefactor A, the stress exponent n, the grain size d, 254 the grain size exponent m, the strain rate $\dot{\epsilon}$, the activation energy E, the activation vol-255 ume V, and the gas constant R. P and T describe pressure and temperature, and the 256 index i refers to either diffusion or dislocation creep. In equation (1), η_i and η_{ref} give 257 the viscosity prefactor for layer i and the reference viscosity used to implement the vis-258 cosity jump underneath the asthenosphere. The exponential terms describe the weak-259 ening effect of melt with melt fraction ϕ (i.e. the porosity field) and exponential pref-260 actor α_{ϕ} , and the strengthening effect of melt depletion via the fraction of melt residue 261 ψ (i.e. the positive peridotite field) and the corresponding exponential prefactor α_{ψ} , which 262 is in this case set to 0 (no depletion strengthening). In analogy to Heyn and Conrad (2022), 263 we simplify this viscosity law to a temperature and composition dependent rheology with 264 $m_i = 0, n_i = 1$ and $V_i = 0$. All values for the parameters are listed in Table 1. 265

Melting and freezing are calculated relative to the dry solidus following Dannberg and Heister (2016), defined by the surface solidus and the pressure gradient given in Table 1, as

$$T_{\rm sol} = T_{\rm sol,0} + \Delta T_p p + \Delta T_\psi \psi \tag{3}$$

Table 1: Characteristic parameters used in the numerical simulations that are referred to in the text. The solidus pressure gradient is given for the different models in the order they are mentioned in the text, i.e. "2 steps stagnant plate", "2 steps moving plate", "1 ramp moving plate", "a step moving plate", and "2 steps symmetric moving plate". To see a full list of parameters, including those remaining at the default value specified in the material model, readers are advised to look at the provided parameters files and material model plugin for ASPECT.

Parameter	Symbol	Value [Unit]
Gravitational acceleration	g	$9.81 \ [m/s^2]$
Reference density	ρ	$3300 [kg/m^3]$
Surface velocity (moving plate cases)	v	$2 [\mathrm{cm/yr}]$
Reference viscosity	$\eta_{\rm ref}$	$1 \cdot 10^{22} [Pa \cdot s]$
Viscosity prefactor for upper/ lower layer	η_j	$5 \cdot 10^{22} / 1 \cdot 10^{24} $ [Pa·s]
Prefactor melt weakening	α_{ϕ}	27
Prefactor depletion strengthening	α_{ψ}	0
Viscosity prefactor	$A^{'}$	$8 \cdot 10^{-12} [1/Pa \cdot s]$
Stress exponent	n_i	1
Grain size exponent	m_i	0
Activation energy	E_i	250 [kJ/mol]
Activation volume	V_i	$0 [m^3/mol]$
Surface solidus temperature	$T_{\rm sol,0}$	1350 [K]
Solidus pressure gradient	ΔT_p	7.8/6.8/9.0/7.8/7.8 ·10 ⁻⁸ [K/Pa]
Solidus depletion change	ΔT_{ψ}	200 [K]

via the surface solidus $T_{\rm sol,0}$, the pressure gradient ΔT_p and the depletion change ΔT_{ψ} , 270 with the the pressure p and the depletion (positive peridotite field) ψ . Hence, melt de-271 pletion increases melting temperatures. Once melting has occurred, active melt and melt 272 residues are tracked via compositional fields termed 'porosity' and 'peridotite'. While 273 porosity tracks the active melt fraction (and is therefore always positive), peridotite refers 274 to the fraction of mantle material that has been affected by melt. Therefore, it can have 275 either negative values (for enriched material representing recrystallised melts) or pos-276 itive values (for melt-depleted material). The migration of melt, which includes heat ad-277 vection, is modelled via two-phase flow according to Darcy's law, assuming that melt moves 278 through pore spaces of the mantle. As a consequence, no magma chamber can form, and 279 the effect of Earth's rotation can be neglected. Instantaneous melt volumes at any given 280 time can be obtained by integrating the porosity field over the domain, and time-integration 281 results in cumulative melt volumes. However, since melting and freezing happen on much 282 shorter time scales than mantle convection, both melt volume calculations based on the 283 porosity field may not be able to capture all melt produced during the model runs, and 284 therefore only provide first-order estimates of melt volumes. Furthermore, modeled melt 285 volumes strongly depend on the surface solidus and the pressure gradient, which are not 286 well constrained and vary between the models to avoid extreme melt fractions. 287

Having described the relevant governing equations and technical terms above, it is important to establish and clarify some definitions. In the following text, melt refers to the active melt fraction at any given time step as obtained from the porosity field. Melt that cools down and freezes is not included. Based on the average melt fraction per model cell and its area, we can calculate the total 'melt area', which is technically not a volume since models are 2-D. To convert this into volume estimates, we assume that the model cells are 1 km deep with no variation of melt fraction along this direction, result-

ing in volumes of "km³ per km assumed model extent". Note, however, that this approach 295 cannot represent real 3-D melt volumes, but is likely to underestimate actual melt vol-296 umes because melting areas are typically larger than 1 km in each direction. Therefore, 297 we use calculated cumulative (time-integrated) and peak instantaneous melt volumes only as comparison between different models (as far as the parameters allow), but do not com-299 pare them with estimates of erupted or intruded melt for HALIP. In addition, these mod-300 els do not 'erupt' the melt onto the surface of the model (e.g. via extrusive volcanism) 301 nor assume any removal of melt as done in other studies that calculate "simplified" melt 302 fractions as postprocessing step (e.g., Ballmer et al., 2011; Bredow et al., 2017; Stein-303 berger et al., 2019) - the difference between these types of model and models with melt 304 migration are discussed in sections 5 and 6. The reason that mantle convection models 305 do not include actual eruptive volcanism, and few include melt migration at all, is that 306 the computational effort needed to bridge the time and length scales of larger scale man-307 tle convection and eruptive modelling is expensive. This study serves therefore as a first-308 order estimate on the spatial and temporal dynamics of plume-lithosphere interactions 309 in the presence of melt migration. For nwo, we simply assume that part of the melt will 310 reach the surface and erupt to form HALIP's extrusive magmatism. 311

³¹² 4 Results of plume-lithosphere interaction with melt migration

The basic plume-lithosphere interaction for melt-free models including their parameter dependence has been discussed in (Heyn & Conrad, 2022). Here, we added more complexity by including melt migration and non-uniform lithosphere thickness to see whether the interaction of lithospheric steps can explain all or part of the multi-pulse melting behaviour and complexity of HALIP. For simplicity, we will first analyse a model with a stagnant plate, before considering moving plate scenarios.

319

4.1 Stagnant plate with 2 lithospheric steps

In order to simulate the transition between a continental edge to a more cratonal 320 interior, i.e. from the Sverdrup Basin to the cratonic parts of Greenland, we implemented 321 two lithospheric steps 500 km apart from each other, increasing the lithosphere thick-322 ness from 100 km in the northwest (Sverdrup Basin) to 200 km underneath the Green-323 land craton in the southeast (see Figure 4a). For the initial condition, the plume is placed 324 underneath the first step (towards the left of the domain, representing the northwest-325 ern part of the transect), such that it interacts with the lithosphere close to the mod-326 elled margin of the continent. The existence of melt at a given time step is indicated by 327 the melt fraction in Figure 4c, which is plotted on top of the temperature field. The first 328 melt appears after about 12 Myr of model time (Figure 4b), when the plume head as-329 cends and reaches a depth of about 200 km. However, initial melt fractions are small and 330 focused in the top area of the plume head (indicated by the black pixels Figure 4c). Within 331 the next 1 - 2 Myr, the plume head reaches the LAB, and tilts to the left/northwest, 332 towards the thinner lithosphere of the Sverdrup Basin, resulting in strongly asymmet-333 ric spreading of the plume material. This asymmetry becomes more pronounced over time, 334 with significantly less plume material spreading below the margin (second step, e.g. vis-335 ible after 21 Myr). 336

In this model, melting only occurs beneath the left/northwestern part of the do-337 main (4c), and its distribution varies significantly with time and space (4b,c). Within 338 just 1 Myr (between 12 and 13 Myr), the melt initially generated in the central portion 339 of the plume head $(12 \,\mathrm{Myr})$ has risen advectively and reached the LAB. This interaction 340 generates significantly more melt by 13 Myr due the heat advected with the rising 341 melt and a lower solidus temperature of the ambient material. Additionally, at 13 Myr, 342 a second melting area to the left/northwest has formed in another branch of the plume 343 head around $100 - 200 \,\mathrm{km}$ away. This is separated from the older central melt region 344



Figure 4: Results for a 2-D model with 2 lithospheric steps and a stagnant plate, with (a) the model setup including domain dimensions, (b) the instantaneous and cumulative melt volumes over time, and (c) selected snapshots of the temperature field (blue-red colours) overlain with active melt fraction (black-magenta-yellow colours). The lithosphere-asthenosphere boundary (LAB, corresponding to the 1400K isotherm) is indicated in (a) and (c) by the dark line. The calculated instantaneous melt volumes in (b) correspond to the integrated melt fractions shown in (c) at the indicated times. Note, however, that 2-D models do not give actual melt volumes, but rather 2-D "melt areas" in km², which are then for simplicity converted to melt volumes assuming an extent of 1 km in the third dimension (see section 3 for more information; a more appropriate term could therefore be "melt volumes per km"). The purple dotted rectangle in (a) marks the zoom-in of the domain shown in (c), and the time-evolution of melt volumes in (b) is compressed after 60 Myr since no more melt is generated in the model. White dashed boxes mark the zoom in for Figure 6.

by a downwelling due to local small-scale convection. The downwelling migrates to the 345 left with time, and is also visible at 21 and 27 Myr. By 21 Myr, the total amount of melt 346 has decreased significantly (indicated by lack of purple/yellow colours in Figure 4c and 347 the melt volume in Figure 4b), with the second melting region being almost melt-free, 348 and the initial melting area being spread out and directly interacting with the LAB. Due 349 to the erosion of the lithosphere by melt-induced small-scale convection, the initial melt-350 ing region experiences rejuvenated strength of melting by about 27 Myr (Figure 2b and 351 4c, lowermost panel). After this episode of rejuvenated melting, the amount of melt in 352 the model subsides, and the model reaches a melt-free state from about 50 Myr. Even 353 though the mantle plume is still active (switched off at 75 Myr), it does not initiate fur-354 ther melting, neither above the plume nor in any of the previous melting regions. For 355 this model, the duration of melting is ca. 38 Myr (from 12 - 50 Myr). 356

4.2 Moving plate with 2 lithospheric steps

As a next step, we introduce a plate moving with a constant velocity of $2 \,\mathrm{cm/yr}$ 358 towards the left/northwest, such that the cratonic part of the model eventually moves 359 towards and over the plume. The initial setup is similar to the case shown in Figure 4a, 360 but the model domain is extended towards the north, with the steps starting 300 km and 361 800 km right/southeast of the plume position (see Figure 5a). The distance to the first 362 step is chosen so that the plume head hits the transition between the basin and the con-363 tinental margin, similar to the stagnant plate case. The first melting occurs at around 365 14 Myr (Figure 5b), and increases significantly until the plume head hits the LAB at 15 Myr, just in front of the first step where the basin transitions to the margin (Figure 5c). The 366 instantaneous melt volume at the peak of this initial pulse is around $520 \,\mathrm{km}^3$ compared 367 to $300 \,\mathrm{km^3}$ in the stagnant scenario, but the cumulative melt volume is much bigger for 368 the stagnant plate case (about $26,800 \,\mathrm{km^3}$ for stagnant plate vs. $15,900 \,\mathrm{km^3}$ for moving 369 plate). The plume and region of melt at this time, and after, is tilted and deflected to-370 wards the left/northwest, towards the thinner lithosphere of the basin. As in the stag-371 nant plate model, the resulting time-dependent melt distribution is spatially inhomoge-372 neous. From ca. 15-21 Myr there is a large amount of melting occurring close to the 373 continental margin (first step), and a second melting region is developing about 500 km 374 away from the step to the left/northwest, underneath the basin. With increasing time, 375 the steps move towards the northwest relative to the plume, and at 21 Myr, the conti-376 nental margin begins to pass over the underlying plume position. 377

At around 35 Myr, the plume material, as indicated by both the thermal and melt 378 fraction fields, covers an area extending more than 1000 km away from the active plume, 379 which is now located under the second step (craton). At this time, there is no more melt-380 ing occurring in the model. However, over time, the topography of the LAB has facil-381 itated the channelling of the shallowest plume material to the left/northwest, towards 382 the thinner lithosphere of the basin. A result of this is that around 45 Myr, plume ma-383 terial arriving at the southeastern edge of the basin in front of the first step is hot enough to melt, causing "rejuvenated" melting underneath the same region of the basin that has 385 previously been affected by melting when the plume head arrived at 15 Myr. This melt-386 ing occurs in two small pockets, separated by a small downwelling, and is related to lo-387 cal small-scale dynamics of plume material interacting with locally thinned lithosphere. 388 Note, however, that the melt fractions and the amount of melt are much smaller than 389 for the first batch of melting (about 0.013 maximum melt fraction and $300 \,\mathrm{km^3}$ cumu-390 lative volume compared to 0.23 maximum melt fraction and $15,600 \,\mathrm{km^3}$ cumulative melt 391 volume, respectively). It is also worth noting that this "rejuvenated" melting in this model 392 (c.f. the second pulse at 27 Myr in the stagnant case) lasts for at most 10 Myr and hap-393 pens approximately 30 Myr after the initial melting (see Figure 5b). As will be discussed 394 later in more detail, both the length, timing and position of rejuvenated volcanism cor-395 respond to the constraints from HALIP data for the first and second pulse of magma-396 tism obtained from Dockman et al. (2018, see also Figure 2), while relative volumes for 397 HALIP are extremely difficult to quantify and may not be represented correctly in the 398 models. Finally, it is noteworthy that the melting at this time occurs more than 500 km 300 away from the active plume stem, indicating that the presence of melt cannot necessar-400 ily be used as a constraint for the central plume position at the given time. The dura-401 tion of melting in this scenario is ca. 21 Myr (from 14 - 35 Myr) or ca. 33 Myr (14 -402 47 Myr) if considering the later, distal episode. 403

404

357

4.3 Lithosphere thinning and its relation to melting regions

As described in Heyn and Conrad (2022) for melt-free models (i.e. neither simplified melt fractions nor migration included), the lithosphere above a mantle plume starts to thin as soon as the plume head reaches the lithosphere-asthenosphere boundary due to thermo-mechanical erosion from the plume. With continued plume-lithosphere inter-



Figure 5: Model setup for a case with 2 lithospheric steps in which the craton moves over the plume with an imposed plate velocity of 2 cm/yr indicated by arrows (a), instantaneous and cumulative melt volumes versus time (b), and temperature field snapshots with porosity for given times (c). As for Figure 4, the rectangle in (a) marks the area of the temperature snap shots shown in (c), and times from the snap shots in (c) are marked for the instantaneous melt volume in (b). The edge of the basin is marked by the small black vertical line, and the dashed white boxes mark the zoom in used for Figure 6. In contrast to the stagnant plate case, this model reaches a melt-free stage at about 35 Myr, before "rejuvenated" melting occurs at around 42 Myr underneath the right/southeastern corner of the basin.

action, the local thinning becomes more pronounced, reaching a maximum value shortly 409 after the plume is removed (for stagnant plate cases) or the respective plume-affected 410 area of the lithosphere has moved significantly (about $200-300 \,\mathrm{km}$) relative to the plume 411 (for moving plate cases). In contrast to the melt-free models of Hevn and Conrad (2022), 412 we find that melt rising from the melting zones in the plume head can further reduce litho-413 sphere thickness locally. Figure 6 plots the amount of lithospheric thinning occurring across 414 the horizontal direction at two selected time steps for the two models with two lithospheric 415 steps as discussed above (c.f. Figures 4a and 5a). These particular time steps were cho-416 sen because they correspond to changes in the melt fraction (e.g. panel b Figure 4 and 417 5). As seen in Figure 6a for the stagnant plate model (corresponding to Figure 4), thin-418 ning starts at 13 Myr above the melting region (rightmost corner of the basin, next to 419 the first step), but the effects are small since the melt has just reached the LAB. The 420 presence of melt, and its ability to rise faster than the rest of the plume, reduces local 421 viscosity, increases local dynamics and therefore causes local erosion while a plume with 422 simplified melt fractions at the same point is starting to spread out at about 50 km be-423 low the LAB (compare Figure 9a). At 27 Myr, the lithosphere is thinnest above the left-424 most part of the prolonged primary melting region, where melting is still active (c.f. Fig-425

ure 4c), while the lithosphere thickness is less affected above the second melting region 426 further left/northwest and the area closer to the step where only a small amount of melt 427 remains.

428



Figure 6: Depth of the LAB (defined as the 1400 K isotherm) at given times for the models with two lithospheric steps discussed above, with the (vertically stretched) temperature fields and melt fractions from Figures 4c and 5c added. While the arrival of melt at the LAB at 13 Myr in (a) only causes a small reduction in thickness, this effect is significantly more pronounced at later times. Small-scale undulations are related to the presence of melt, and are therefore absent underneath the thicker parts of the lithosphere where no melting occurs.

A similar observation can be made in Figure 6b for the moving plate case (corre-429 sponding to Figure 5). At 21 Myr, melting is still active the in the primary melting zone 430 close to the step, where the lithosphere is significantly thinner than it is further away 431 from the plume. The melting region further to the left/northwest (which did not reach 432 the LAB in Figure 5c at 21 Myr) seems to have little effect on lithosphere thickness at 433 this particular time step. After melting ceases, the lithosphere starts to heal and increases 434 in thickness, as expected. Yet, some of the melt-induced undulations of the LAB remain, 435 and local dynamics can reactivate these areas at 45 Myr. Even though melting itself hap-436 pens at about 125-150 km depth, the presence of a thinned lithosphere enables local con-437 vection that brings plume material up into the melting zone. Note, however, that the 438 local thinning of the lithosphere by melt is limited both in space and time, and reflects 439 the amount of melt that was locally present. As a consequence, the lithosphere beneath 440 the thicker continental margin or the craton to right of the plume is significantly less af-441 fected than the lithosphere beneath the basin, and LAB undulations exhibit much longer 442 wavelengths and lifetimes. Since surface heat flux is a time-delayed and long-wavelength 443 filtered image of the lithosphere thinning (Heyn & Conrad, 2022), it does not reflect the 444 deeper, local, time-dependent melt distributions as predicted in these models. However, 445 if melt were to migrate significantly closer to the surface in these models, then it would 446 exert a larger (but still local) effect on surface heat flux. 447

448

4.4 Influence of lithospheric structural variations

In order to investigate the robustness of the rejuvenated volcanism for models with 449 a moving plate, we tested three alternative initial LAB topographies; a gradual ramp, 450

a single step, and an indentation set-up. For the first alternative model, we followed the 451 conceptual model of Dockman et al. (2018), and included a gradual increase of lithosphere 452 thickness from 50 km underneath the Sverdrup basin to 100 km underneath the Green-453 land craton (see Figure 7a), starting initially 300 km right/southeast of the plume. Due to the reduced lithosphere thickness, we increased the pressure gradient for the solidus 455 to reduce overall melt volumes (see Table 1), while all other parameters are kept con-456 stant. As for the case of 2 discrete lithospheric steps (Figures 4 and 5), the initial melt-457 ing zone in the plume head quickly separates into separate regions (here three), sepa-458 rated by local downwellings $(15 \,\mathrm{Myr}, \mathrm{Figure} \, 7\mathrm{c})$. The peak of melt is the highest for any 459 model, reaching about $700 \,\mathrm{km^3}$ at ca. 15 Myr, and the cumulative melt volume of about 460 $28\,600\,\mathrm{km}^3$ is also the largest of all models. In contrast to the previous case, the plume 461 head is initially less tilted, and one branch of the plume head with melting forms under-462 neath the lithospheric ramp, resulting in a total of three distinct melting zones. Hence, 463 the larger peak of instantaneous melt volume (also reflected in the largest melt fraction 464 of (0.31) and larger cumulative melt volume may partly be related to reduced overall litho-465 sphere thickness, despite the increased pressure gradient of the solidus, and partly to the 466 flatter LAB topography that allows for a third melting zone (instead of two as in the pre-467 vious cases). At 21 Myr, the two side branches of melting are stretched significantly and 468 have a larger distance from the central melting zone, while showing only marginal amounts 469 of melt (as indicated by only black pixels). The central melting zone at the base of the 470 ramp (right/southeastern edge of the basin) is no longer above the plume, but still main-471 tains a melt fraction of about 0.15, and instantaneous melt volumes reach a small sec-472 ondary peak (see Figure 7b). However, with the plate moving further to the left/northwest, 473 melting stops completely and the model becomes melt-free (30 Myr), before the central 474 melting region becomes re-activated around 42 Myr. As before, this rejuvenated melt-475 ing happens about 25 Myr after the initial melting, with much smaller melt volumes (\sim 476 $250\,\mathrm{km^3}$ cumulative melt volume) and fractions (~ 0.014) forming over a duration of 477 a few Myr (Figure 7b) and about 600 km away from the plume stem. Hence, multiple 478 melting events can be produced with different initial LAB topographies, as long as the 479 LAB channels plume material towards a thinned area, for example a region that has been 480 influenced by melt before. 481

Results from two additional model setup scenarios (a single step or 2 steps sym-482 metric around a thinned area, Figure 8) show peak instantaneous melt volumes that are 483 smaller than for the cases discussed above. In these cases, there is not a clear sign of re-484 juvenated volcanism. Note, however, that for both of these cases the plume stays longer 485 beneath the basin and the pressure gradient of the solidus temperature is increased com-486 pared to the case with 2 consecutive steps (see Table 1; the value used is the same as for 487 the stagnant plate case) in order for these models to run without generating > 50% melt 488 fraction, which would make the model numerically unstable. For the single step case (Fig-489 ure 8b; red line), the lithospheric step to the craton is 500 km away from the arriving plume 490 head, and the extended basin allows the plume material to spread more evenly and eas-491 ily towards both sides, resulting in an instantaneous melt volume above the plume of about 492 $310 \,\mathrm{km^3}$ at 16 Myr. Due to the smaller initial melt volume, the lithosphere is less thinned, 493 and it becomes more difficult to re-activate any melting zone once the plume interacts 494 with the step (craton). In addition, the lithosphere has more time to heal before the plume 495 material is channeled to the left/northwest by the craton. For the model with a gap (two 496 symmetric steps), it may be expected that plume material rising into the gap of conti-497 nental lithosphere will be trapped there. In this case, prolonged melting periods may be 498 expected, and perhaps also rejuvenated magmatism once the plume moves underneath 499 the continental lithosphere on the other side of the gap. However, our model (Figure 8b; 500 black line) does not show this, potentially because the initial melt volume in the plume 501 head is too small ($\sim 240 \,\mathrm{km^3}$ at 19 Myr) to cause enough lithosphere thinning for re-502 juvenated melting. 503



Figure 7: Analogous to Figure 5, but for a model with a gradual transition between the basin and the craton instead of two steps, with the edge of the margin marked by a black vertical line. As before, the imposed velocity is 2 cm/yr, as indicated by the arrows in the model setup (a), and the time evolution of melt volumes is focused on the first 60 Myr (b). As shown in the snap shots of the temperature field (c), the initial melting zone is split into 3 separate branches at 15 Myr, but only the central branch can sustain substantial melting at 21 Myr. At 30 Myr, the model is melt-free, but experiences rejuvenated melting around 41 Myr underneath the right/southeastern part of the basin.

Finally, we compare the results of our cases with variable LAB depth to a case with 504 a moving plate and a constant lithospheric thickness of 100 km, i.e. a model without any 505 cratonic or continental parts interacting with the plume ("uniform LAB" in figure 8a). 506 The pressure gradient of the solidus for this case is set to $7.6 \cdot 10^{-8}$ K/Pa, and instanta-507 neous melt volumes for this model are shown in Figure 8a as the dashed blue line. No 508 second pulse of melting occurs, despite the fact that the instantaneous melt volume as-509 sociated with the arrival of the plume head $(\sim 950 \,\mathrm{km^3})$ is even larger than for the two 510 lithospheric steps or the gradient case, and an overall duration of magmatism of 20 Myr. 511 Since there is no thicker part of the lithosphere moving over the plume, less plume ma-512 terial is directed to a previously thinned area of lithosphere, and melting zones will not 513 be reactivated. Hence, both LAB topography and sufficient initial melting seem to be 514 necessary to generate rejuvenated magmatism. 515

⁵¹⁶ 5 Comparison of melt migration versus "simplified" melt fractions

As far as we are aware, and given our model setup, this work is the first time that rejuvenated melting, unrelated to plume flux changes, has been observed in numerical models. Previous work looking at plume-lithosphere interactions and the emplacement of large igneous provinces (e.g., Steinberger et al., 2019; Duvernay et al., 2022) used "sim-



Figure 8: Comparison of instantaneous melt fractions for different initial LAB topographies with a zoom-in to the rejuvenated magmatism (a), and the model setups for a case with one step (b), and 2 symmetric steps (c). Note that the rejuvenated magmatism is only visible for the two lithospheric steps (purple line) shown in Figure 5 and the gradual increase in lithosphere thickness (yellow line) from Figure 7. However, peak instantaneous melt volumes are smaller for the model setups with one step (red line) and two symmetric steps (black line), potentially reducing the chance to observe rejuvenated melting. On the other hand, a model with constant lithospheric thickness (dashed blue line) has the largest maximum instantaneous melt volume of all models shown here, but does not show rejuvenated magmatism. Hence, a large instantaneous melt volume alone seems to be insufficient to generate rejuvenated magmatism if LAB topography is absent.

plified" melt fractions instead of melt migration. Simplified melt fractions are calculated 521 as a post-processing step based on the temperature field, and it is typically assumed that 522 all melt generated at a given time is immediately extracted from the system via volcanic 523 eruptions. In contrast, melt migration involves solving for two-phase flow and the ac-524 tual migration of material (melt/residue) and heat - the advantage of this being a much 525 more realistic feedback between melt and local dynamics. On the other hand, the sim-526 plified melt fraction approach automatically generates estimations of erupted melt vol-527 umes, while the melt migration method does not dynamically erupt any melt and would 528 therefore require the setting of an appropriate threshold to estimate erupted melt vol-529 umes. Using melt fractions, the study of Steinberger et al. (2019) investigated how the 530 North Atlantic Igneous Province (NAIP) may have been emplaced by the Iceland Plume 531 interacting with the Greenland craton, and was able to reproduce the reported time frames 532 and melt distributions around Greenland. In contrast, Duvernay et al. (2022) took a more 533 general approach and tested how different settings of lithosphere thickness affect the gen-534 eration and patterns of simplified melt fractions. Their results show that melt distribu-535 tion can vary significantly depending on where the plume impinges on the lithosphere, 536 resulting in a very inhomogeneous distribution of the inferred magmatism over time and 537 space. However, neither of their models reproduced rejuvenated volcanism, as has been 538 suggested for HALIP. Based on our models, we suggest that the missing component is 539 the local small-scale dynamics of the melt, which can only be captured by modelling full 540 melt migration. The dynamic feedback between mantle convection and melt is necessary 541 to thin the lithosphere locally enough to facilitate rejuvenated melting, which would oth-542 erwise require significant changes in plume strength or lithospheric rifting. 543



Figure 9: Comparison of 2-D models that implement simplified melt fraction (top row, (a) and (b)) with those that include 2-phase flow melt migration (bottom row, (c) and (d)). Models feature 2 lithospheric steps and either a stagnant plate (left panels (a) and (c), see Figure 4) or a moving plate (right panels (b) and (d), compare Figure 5), and melt is shown on top of the temperature field. Velocities are shown as white arrows, with the scale giving the maximum velocity in each case. While maximum melt fractions are similar in all cases, the melting zones for simplified melt fractions are much broader and more homogeneous than for the cases with melt migration. Note also that plume-related melt in the melt migration cases (bottom row) can rise to shallower depths (compared to the top row, where melting is confined within the plume head), effectively transporting heat to shallower depths.

As mentioned, the calculation of simplified melt fractions happens as a post-processing 544 step, hence the presence of melt does not influence the dynamics. Figure 9 compares mod-545 els with simplified melt fractions (top panels) to models that include melt migration for 546 two setups: one with a stagnant plate (Figure 9a,c) and one with a moving plate (Fig-547 ure 9b,d), both with two lithospheric steps. For both the stationary and moving mod-548 els that only calculate simplified melt fractions (Figure 9a,b), the result is a more ho-549 mogeneous spread of melting across the plume head. The uppermost edge of the plume 550 head, as indicated by the pattern of the melt fraction field, is more or less parallel to the 551 LAB. Furthermore, the melting zone occupies most of the upper third of the plume head, 552 with the largest melt fractions (yellows) in the upper part of the melting zone. The cor-553 responding mantle flow field shows rising material in the plume stem, which spreads par-554 allel to the lithosphere underneath the basin region, with only a small portion of the plume 555 moving underneath the first lithospheric step to the right/southeast (continental mar-556 gin). In contrast, melt migration models (lower panels) show two separated or loosely 557 connected melting zones (as described in the above section). Melt generation and dis-558 tribution are very inhomogeneous, with large melt fractions being much more localized 559 than for the models that exclude melt migration. These local dynamics are driven by 560 density and viscosity variations introduced by melt as it evolves through time. This is 561 also reflected in the velocity field; there is increased flow speed and local upwelling in 562

regions where larger amounts of melt are being generated. For example, for the moving plate scenario, the maximum velocity is 20 cm/yr for the simplified melt fraction case and 51 cm/yr when including melt migration. In turn, these localized upwellings cause adjacent downwellings, splitting the melting zones and resulting in a small convection cell next the lithospheric step. Maximum melt fractions (i.e. reaching up to 0.25) at this particular time step are comparable between models with and without melt migration, but total volumes of melt differ significantly due to the smaller melting zones with melt migration.

571 The presence of local dynamics visible in Figure 9 for models with melt migration makes these models significantly more dynamic and time-dependent than models with-572 out melt migration. Since melt affects both density, viscosity and temperature, the equa-573 tions are coupled in a non-linear way, and these models require more computation time 574 and are more sensitive to the chosen parameters. Furthermore, when melt migrates across 575 the domain, its excess temperature and effect on further melting can cause a sort of pos-576 itive feedback, resulting in increasing melt volumes and fractions being generated in a 577 localized area; an effect that cannot be observed when calculating melt fractions as post-578 processing step. As a result of these local dynamics, both lithospheric thinning and mag-579 matism will be more variable in time and space when melt migration is included. How-580 ever, our models cannot simulate volcanism or dyke emplacement, since this would re-581 quire even finer spatial and temporal resolution. Yet, including the local dynamics of melt 582 seems to be necessary when modelling complex scenarios of plume-lithosphere interac-583 tion such as the emplacement of HALIP (compare Figures 1 and 2). While the assump-584 tion of melt being immediately extracted to the surface and not having any significant 585 dynamic influence on mantle and lithosphere dynamics may be reasonable for settings 586 with thin oceanic crust where melt can easily reach the surface, a thicker continental or 587 cratonic lithosphere may require a more sophisticated approach that includes melt mi-588 gration. A thick lithosphere can reduce the fraction of melt that can be extracted from 589 the system, potentially even causing larger amounts of melt to pond beneath the litho-590 sphere for a period of time (e.g., Aulbach et al., 2007, 2017; Sun et al., 2020). In this 591 case, melt will have enough time to interact and affect local dynamics, which should there-592 fore be taken into account when doing numerical models. 593

594 6 Discussion

Two of the main particularities of HALIP are the apparent extensive timing of em-595 placement (over 50 Myrs) and the documented pulses (e.g. earliest magnatism dating 596 at 131 Ma, and pulses at 122, 95, 81 Ma Tegner et al., 2011; Dockman et al., 2018). These 597 observations are in strong contrast to most other LIPs around the world, which are erupted 598 within a very short time scale. As a consequence, there is an ongoing discussion as to 599 whether HALIP is a large igneous province or not. Alternative explanations for at least 600 part of the volcanism have been proposed, for example edge-driven convection north of 601 the Greenland craton as the source for the secondary pulses (Dockman et al., 2018). Al-602 though edge-driven convection is known to be present at lithospheric steps (e.g., Manjón-603 Cabeza Córdoba & Ballmer, 2021, and references therein), it has been shown that the 604 amount of melt being generated by this mechanism is small compared to plume-induced 605 melting (Manjón-Cabeza Córdoba & Ballmer, 2021, 2022), and may only sustain small 606 volcanic features such as seamounts. Another mechanism to generate melt in the pres-607 ence of lithospheric steps is shear-driven upwelling (Conrad et al., 2010), but as for the 608 edge-driven convection, expected melt volumes are small. So far, it has not been shown 609 numerically, or otherwise, that edge-driven convection or shear-driven upwelling alone 610 can produce melting as observed for HALIP. In contrast, our numerical models of a ther-611 mal mantle plume interacting with LAB topography dynamically produce rejuvenated 612 magmatism with comparable timing and duration as has been observed for HALIP. While 613 an extended suite of model runs with alternative parameter setups is beyond the scope 614

of this study, our results show that a mantle plume-related origin for many of the ob-615 served HALIP pulses is possible. Furthermore, HALIP magnatism shows compositions 616 and characteristics of plume influence (e.g., Tegner et al., 2011; Buchan & Ernst, 2018; 617 Bédard, Troll, et al., 2021; Senger & Galland, 2022). That said, depending on the paleo-618 topography of the LAB, edge-driven convection or shear-driven upwelling (in addition 619 to the convective patterns modelled here) may contribute to local dynamics and over-620 all melt volumes and distributions (e.g., Conrad et al., 2010; Manjón-Cabeza Córdoba 621 & Ballmer, 2022; Duvernay et al., 2022; Negredo et al., 2022). 622

623 Previous studies looking at melting in plumes have focused on different aspects of the problem, including melting in thermochemically zoned plumes (e.g., Dannberg & Gassmöller, 624 2018), melting in the presence of continental or cratonic lithosphere (Duvernay et al., 625 2022), or melting for specific plumes and tectonics settings (e.g., Ballmer et al., 2011; 626 Bredow et al., 2017; Steinberger et al., 2019; Liu et al., 2021; Manjón-Cabeza Córdoba 627 & Ballmer, 2022; Negredo et al., 2022). Even though the various model setups and lev-628 els of complexity are different, all studies of plume dynamics mentioned above have in 629 common that they only consider simplified melt fractions, and do not model melt mi-630 gration, as we do in this study. Some studies vary parameters related to melting changes 631 such as the density (Ballmer et al., 2011; Bredow et al., 2017; Steinberger et al., 2019; 632 Liu et al., 2021; Manjón-Cabeza Córdoba & Ballmer, 2022), viscosity (Ballmer et al., 2011; 633 Bredow et al., 2017; Steinberger et al., 2019; Liu et al., 2021), or melting temperature 634 (Ballmer et al., 2011; Negredo et al., 2022), and the work of Ballmer et al. (2011) on the 635 Hawaiian Plume shows rejuvenated volcanism next to the plume track due to small-scale 636 convection. While the role of small-scale convection is important in both the work 637 of Ballmer et al. (2011) and this study, the cases are not directly comparable. Hawaii 638 is located far from any continent or craton on oceanic lithsphere older than $\sim 70 \,\mathrm{Myr}$, 639 which may have developed a washboard pattern of LAB topography before the plume 640 hit the lithosphere (Ballmer et al., 2011). Rejuvenated melt is then generated by inter-641 action of spreading plume material with this pre-existing pattern, but this is not appli-642 cable to the Arctic and HALIP. Furthermore, rejuvenated melting in Hawaii seems to 643 happen within about 10 Myr, while HALIP volcanism is spread out over more than 30 Myr. 644

All of the studies given above assume that melt is immediately extracted from the 645 model, and thus melt volumes are estimated as a post-processing step (Ballmer et al., 646 2011; Bredow et al., 2017; Steinberger et al., 2019). As a result, most of these studies 647 show broad melting areas as we obtain for melt fractions in Figure 9 (top panels). This 648 broad pattern contrasts the localized melting zones that evolve with melt migration (Fig-649 ure 9 lower panels). Since most of previous studies look at plumes that impinge on oceanic 650 lithosphere, it seems likely that most of the melt is quickly erupted, and thus the effect 651 of excluding melt migration might be small. In contrast, continental or cratonic litho-652 sphere may pose a barrier to rising melt, especially if the craton is intact (Aulbach et 653 al., 2017). We have shown that if a significant portion of the melt will remain in the as-654 thenosphere or lower lithosphere, it can be expected that the presence of melt signifi-655 cantly alters the local convection patterns (e.g. Figure 9). This feedback between melt 656 and local dynamics is the cause of the rejuvenated magmatism we see in this study, and 657 may explain the main difference between this work and the study of Duvernay et al. (2022), 658 who report complex melting patterns in the presence of cratonic lithosphere, but do not 659 observe rejuvenated volcanism. However, to capture plume-lithosphere dynamics correctly 660 in settings with strong LAB topography, models need to include melt migration. 661

Despite the finding that our models can reproduce the prolonged melting and the pulses seen for HALIP, there are several assumptions and limitations that have to be taken into account. First of all, melt migration in ASPECT is modeled via Darcy's law, assuming that melt moves through the pore space of the ambient mantle or lithosphere. As a consequence, we cannot model melt eruptions, LIP emplacement or dyke intrusions, which are beyond the scope of this work. In fact, most of the melt in the models recrys-

tallizes close to the LAB instead of penetrating far into the lithosphere. Hence, our mod-668 els may overestimate the impact of melting on lithosphere thinning. Furthermore, the 669 amount of melt generated within the plume head is strongly dependent on parameters 670 such as the surface solidus and the pressure gradient of the solidus. In addition, we only 671 considered melting of dry peridotite, while the presence of water would facilitate more 672 voluminous melting, or melting at lower temperatures or greater depths. In additon, our 673 models are 2-D, and therefore cannot capture the full dynamics of plumes, which are 3-674 D features. However, models with melt migration are computationally expensive, mak-675 ing 3-D models difficult to realise. 2-D models also do not allow us to properly estimate 676 melt volumes, which is the reason why we do not compare absolute melt volumes to es-677 timates for HALIP. This problem is perhaps mitigated here because it is difficult reli-678 ably estimate melt volume for HALIP directly due to the large spatial spread of volcan-679 ism and difficulties to map intrusive and extrusive magmatism (Tegner et al., 2011; Sen-680 ger & Galland, 2022). Furthermore, calculating time-integrated melt volumes from the 681 porosity field alone is likely to underestimate the total melt volume of the model because 682 melting / freezing and melt migration happen on timescales smaller than the timesteps 683 at which the porosity field is updated. 684

Although modelled melt volumes cannot be compared to observations directly, it 685 is obvious that the rejuvenated magmatism in our models is significantly smaller and more 686 regionally confined than for the initial melting in the plume head (e.g. Figures 5a and 687 7b). For HALIP, it is extremely difficult to estimate the distribution and volumes of in-688 trusive and extrusive magmatism, both for the initial phase and subsequent pulses (Fig-689 ure 2). An areal estimate of $80.000 \,\mathrm{km}^2$ was proposed for the late-stage ($85-60 \,\mathrm{Ma}$, third 690 pulse) alkaline volcanism alone as defined by Tegner et al. (2011). More recently an ac-691 cumulated magma volume of $100,000 \,\mathrm{km^3}$ was proposed by Saumur et al. (2016) for the 692 Sverdrup Basin, without discriminating between the timing of this magmatism. Based 693 on field mapping, Senger and Galland (2022) calculated 0.14–2.5 km³ of emplaced magma 694 in Svalbard alone, and expanding that to a regional (including Barents and Franz Josef 695 Land) time-accumulated magma volume implies up to 200,000 km³ based on geophys-696 ical data. These estimates do not include the ca. $1.3 \cdot 10^6 \text{ km}^2$ areal extent of the Al-697 pha Ridge Magmatic High (HAMH, Oakey and Saltus (2016); Figure 1), which would 698 equal about $200 \cdot 10^6 \text{ km}^3$ magma and therefore far exceed the continental (onshore or 699 continental shelf) estimates listed above. Cumulative melt volumes for HALIP may there-700 fore be in the order of $(200 - 300) \cdot 10^6 \text{ km}^3$, but we lack clear constraints on the vol-701 umes of individual pulses. Hence, it seems plausible that the rejuvenated volcanism we 702 see in our models can explain at least some of the locally confined alkalic magmatism 703 observed for HALIP, but it is difficult to asses whether our model prediction of secondary 704 pulses being about 2 orders of magnitude smaller than the initial pulse is realistic or not. 705 Deviations between modeled and observed melt volumes may be partly explained by the difference between 2-D and 3-D geometry, which would allow for more local patches of 707 melting and more realistic melt volumes. In addition, there are several potential mech-708 anisms that could strengthen rejuvenated melting to produce a larger second pulse or 709 even a third pulse of magmatism in the models. As geological data shows, the Arctic un-710 derwent extension at around the time of HALIP (e.g., Tegner et al., 2011). If melt-thinned 711 areas are affected, this would enhance rejuvenated melting there. Distance and respec-712 tive timing of the extension/ spreading would in this case determine the magnitude of 713 melt generation, hence better constraints on plate reconstructions of the Arctic could 714 be essential to explain HALIP. Another option to generate a more voluminous second 715 pulse of melting could be a temporal variation in plume flux, which has been suggested 716 for the Iceland plume based on V-shaped ridge segments in the North Atlantic (e.g., Ito, 717 718 2001; S. M. Jones et al., 2002; S. Jones et al., 2014; Parnell-Turner et al., 2014). Dynamically, such a plume pulse could be related to (potentially slab-driven) dynamics at the 719 core-mantle boundary (e.g., Heyn et al., 2020), interaction with edge-driven convection 720 (e.g., Manjón-Cabeza Córdoba & Ballmer, 2022; Negredo et al., 2022), solitary waves 721 (Ito, 2001), or an interaction between the plume and the slab found underneath Green-722

land at about 1000-1600 km depth (Shephard et al., 2016). Strong plume pulses or successful seafloor spreading can generate melting without the mechanism described in this
 paper, but a previously melt-influenced region of the lithosphere is more prone to rejuvenated melting, even without small changes in plume flux or failed rifting.

It is beyond the scope and capability of our models to predict compositions of gen-727 erated melts, but our models show that the whole asthenosphere underneath the Sver-728 drup Basin region is influenced by the plume. As a consequence, we would expect that 729 melting zones are plume-fed, with all of the generated melt having more or less plume-730 731 influenced compositions. However, the composition of melts is expected to change over time as the plume-lithosphere interaction evolves, and could also be influenced by pre-732 existing mantle chemistry - whether sub-continental or metasomatic mantle. Initial melts 733 are likely to dominantly have the original (potentially deep-mantle) signature of the plume, 734 which could shift towards enriched mantle (EM) signatures at later stages when more 735 lower lithosphere is entrained in local convection cells where melt is generated. During 736 the evolution of the system, further influences may come from pulses in plume flux, whether 737 derived from the transition zone, lower mantle or plume-slab interactions. Finally, melt 738 fractions and melting depths change significantly during the evolution of the system in 739 our models (especially for the models accounting for melt migration). The first phase 740 of melting in the the plume head is dominated by deep melts at lower melt fractions (that 741 may never reach the surface), which rapidly changes towards predominantly shallower 742 melting with large melt fractions. Over the next few million years, melt is generated at 743 (locally increasingly) shallower depths, but melt fractions decrease until melting stops 744 altogether. Rejuvenated melting is then characterized by low melt fractions at slightly 745 deeper depths than the late stage of the first pulse. As a consequence, magmas can be 746 expected to shift from mostly tholeiitic around the peak of the first pulse, to more al-747 kalic at later stages of the first peak and in the second and third pulses. This trend seems 748 to be supported by the data for HALIP (Figure 2). Models also indicate that the dis-749 tribution of tholeiitic and alkalic magmatism may be not only time- but also location-750 dependent, and both types may be present at the same time, similar to what HALIP mag-751 mas suggest (Figure 2). However, in order to have a better constraint on the timing and 752 locations of each magma type, we would need to add additional complexities to the model, 753 e.g. run 3-D models and include more complex melting laws that track compositions in 754 more detail. Future models could also address the broader HALIP geographic setting 755 (Figure 1), beyond that of the Canadian Arctic Islands focused on here. 756

757 7 Conclusions

HALIP is an unusually large igneous province, if it is a LIP in the classic sense at 758 all, because magmatism lasted for more than 50 Myr, with pulses of volcanic activity at 759 around 122 Ma, 95 Ma and 81 Ma. So far, no conclusive mechanism has been proposed 760 to explain this behaviour, since LIPs are typically emplaced within a short time, and edge-761 driven convection alone seems unlikely to explain the extent and volumes of magmatism 762 in the Arctic, even for secondary pulses (Manjón-Cabeza Córdoba & Ballmer, 2021). In 763 this work, we show that a mantle plume can produce prolonged melting periods and mul-764 tiple events of melting in the same area if a plume interacts with tectonically inherited 765 spatial variations in LAB depth, and if the lithosphere is locally thinned due small-scale 766 convection associated with melt migrating upwards. More specifically, the plume head 767 has to impinge on the thinner lithosphere of the basin, before the thicker continental mar-768 gin and craton move over the plume stem. This variation in thickness channels plume 769 flow in the asthenosphere towards the previously thinned lithosphere of the basin, where 770 rejuvenated melting can occur. Considering the tectonic history of the Arctic and the 771 presence of cratons around it, a non-uniform LAB thickness across the region seems plau-772 sible at the time of plume impingement. Hence, our models argue for a plume-related 773 origin of most or even all the magmatism associated with HALIP. However, geodynamic 774

models investigating HALIP volcanism (and potentially other continental or cratonic LIPs)
 should include 2-phase flow melt migration to more accurately capture the dynamics af fecting melt generation.

As shown by the numerical models, a plume head arriving underneath the extended 778 lithosphere of the Sverdrup Basin would cause melting and further lithosphere thinning 779 with strong local undulations in LAB depth, which can facilitate rejuvenated melting 780 underneath the basin about 25-30 Myr later when the plume is underneath the Green-781 land craton and plume material is channeled towards the thinner basin. Melting depths 782 and melt fractions vary throughout the time, and inferred compositions would be expected 783 to change from more tholeiitic magmas for the first peak towards more alkalic magmas 784 for later stages of the first peak and secondary pulses, but our models cannot track this 785 in detail. While the simplified 2-D model used in this study can explain both timing and 786 duration of the second pulse of melting for HALIP, relative melt volumes may not be rep-787 resentative for HALIP, and a third pulse cannot be reproduced. However, the magni-788 tude of the second pulse of melting, and the presence and size of a potential third pulse, 789 may be affected by contemporaneous tectonics (especially further extension or rifting), plume pulses generated at the core-mantle boundary, or plume-slab interaction related 791 to the slab underneath Greenland (Shephard et al., 2016), all of which could easily re-792 activate or strengthen rejuvenated melting in areas that have been thinned by melt be-793 fore. Finally, our models show that active melting does not necessarily happen above the 794 plume, but lateral flow of plume may re-activate local melting in a previously plume-affected 795 region up to a few hundred km away from the current plume position. Such complex-796 ity should be taken into account when using plume-related melting to infer plume po-797 sitions, and when interpreting patterns of magmatism for HALIP and other LIPS or melting events close to continental or cratonic margins. 799

800 8 Open Research

Data presented in Figures 1 and 2, together with the ASPECT parameter files, AS-PECT material model plugin, model data and postprocessing scripts used to generate and analyse the numerical models presented in this paper, are attached as zip folder for review and will be made available via Zenodo upon acceptance of the paper. ASPECT v2.4.0 (Bangerth et al., 2022) is freely available under the GPL v2.0 or later license, and can be accessed via https://geodynamics.org/resources/aspect or https://aspect.geodynamics.org, or via the github page https://github.com/geodynamics/aspect that includes the current development version of the code.

809 Acknowledgments

The authors acknowledge support from the Research Council of Norway through its Cen-810 ters of Excellence funding scheme, Project Numbers 223272 and 332523, through the MAG-811 PIE project, Project Number 288449, and through its Young Research Talent scheme 812 for POLARIS - Evolution of the Arctic in Deep Time, Project Number 326238. Com-813 putations were made possible by the Norwegian Research Infrastructure Services (NRIS/sigma2) 814 via allocations NN9283K, NS9029K and NN10003K. The authors declare no competing 815 financial interests. We would like to thank the Computational Infrastructure for Geo-816 dynamics (geodynamics.org), which is funded by the National Science Foundation un-817 der award EAR-0949446 and EAR-1550901 for supporting the development of ASPECT. 818

819 References

 Albers, M., & Christensen, U. R. (1996). The excess temperature of plumes rising from the core-mantle boundary. *Geophysical Research Letters*, 23(24), 3567– 3570. Retrieved from https://agupubs.onlinelibrary.wiley.com/doi/abs/

823	10.1029/96GL03311 doi: 10.1029/96GL03311
824	Aulbach, S., Griffin, W. L., Pearson, N. J., O'Reilly, S. Y., & Dovle, B. J. (2007).
825	Lithosphere formation in the central slave craton (canada): plume subcretion
826	or lithosphere accretion? Contributions to Mineralogy and Petrology, 15/(4).
827	409-427, doi: 10.1007/s00410-007-0200-1
020	Aulbach S. Massuveau M. & Gaillard F. (2017) Origins of cratonic man-
920	tle discontinuities: A view from petrology geochemistry and thermo-
029	dynamic models Lithos 268-271 364-382 Betrieved from http://
030	www.sciencedirect.com/science/article/nii/S0024493716303905 doi:
031	10 1016/i lithos 2016 11 004
032	Ballmer M D. Ito C. van Hunen I. & Tackley P. I. (2011) Spatial and tempo-
833	rel veriability in havaiian hotspot volcanism induced by small-scale convection
034	Nature Geoscience $1/7$ 457–460 doi: 10.1038/ngeo1187
000	Bangarth W Dannharg I Fratars M Casemaallar B Clarum A Haistar
836	T Naliboff I (2022 July) Accest $v^{0} / 0$ Zanada Batriavad from
837	https://doi org/10.5281/zonodo 6003424 doi: $10.5281/zonodo 6003424$
838	Pádard I H. Saumur P. M. Tornar C. Troll V. P. Dooran F. M. Evonshield
839	C A Derving K (2021 06) Coochemical Systematics of High Aratic
840	Large Imaging Dravinge Continental Theleijtes from Canada – Fuidence for
841	Diarge igneous r fovince Continental i noientes from Canada—Evidence for Progressive Crustel Contamination in the Plumbing System – <i>Journal of Patrol</i>
842	62(0) argh $0/1$ Betrieved from https://doi.org/10.1002/petrol.org/
843	agab041 doi: 10.1003/petrology/ogab041
844	Bédard I H. Troll V. R. Deoran F. M. Terner, C. Saumur, B. M. Evenebick
845	C A Dewing K (2021 06) High Aretic Large Igneous Province Alle
846	line Dealer in Canada, Evidence for Multiple Mantle Components
847	inte Rocks in Canada: Evidence for Multiple Mantle Components. $Journal$
848	of <i>Fetrology</i> , 02(9), egab042. Retrieved from https://doi.org/10.1093/
849	Prodom E. Steinbargen D. Coggmällen D. & Donnham L. (2017) Herr plune
850	bredow, E., Steinberger, D., Gassmoner, R., & Dannberg, J. (2017). How plume-
851	Fige interaction snapes the crustal thickness pattern of the relimon hotspot truck C_{coch} emission C_{coch} emission $L^{(2)}$ 2020 2048 Detrived
852	from https://orupuba.orlinelibrory.viley.com/doi/oba/10.1002/
853	20170006875 doi: 10.1002/20170006875
854	20176000875 doi: $10.1002/20176000875$
855	ated with the high angle large ignorus province (helip) Conducan Recognition
856	58 20 57 Betrieved from https://www.aciencedirect.com/acience/
857	orticlo/mii/G1242027X18200455_doi: 10.1016/j.gr.2018.02.006
858	Pádard I H Troll V P. Desgan F. M. Terner C. Sauraur P. M. Evensbiek
859	C A Dewing K (2021 06) High Aretic Large Igneous Province Alle
860	line Rocks in Canada: Evidence for Multiple Mantle Components
861	af Petrology $6^{9}(0)$ or $b^{1/2}$ Botrioved from https://doi.org/10.1093/
802	netrology/egsh042 doi: 10.1003/netrology/egsh042
803	Coffin M E $[r Eldholm O (1004)]$ Large ignored provinces: Crustel structure
864	dimensions and external consequences — <i>Reviews of Coordinates 29(1)</i> , 1-36
865	Botrioved from https://agupubs.onlinelibrary.uiley.com/doi/abs/
800	10 1029/93BC02508 doi: https://doi.org/10.1020/03BC02508
807	Conred C P Wu B Smith F I Biance T A fr Tibbette A (2010)
868	Shoar driven unwelling induced by lateral viscosity variations and acthono
870	spheric shear. A mechanism for intraplate volcanism
87U	Earth and Planetary Interiors 178(2) 169–175 Retrieved from https://
871 972	WWW sciencedirect com/science/article/nii/S003102010000200V doi:
δ/2 972	10 1016/i neni 2009 10 001
813	Corfu F Poltonu S Planko S Faloido I I Svenson H Zavensheelt A Pr
874	Stolboy N (2013, 06) U-Ph geochronology of Crotacoous magnetism on
875 976	Svalbard and Franz Josef Land Barants Sea Largo Janoous Province - Coolegi
870 077	cal Magazine 150(6) 1127 1135 Rotrioved from https://doi.org/10.1017/
011	$c_{w} muguzine, 100 (0), 1121-1100. Therefore non-interpendent of g, 10.1017/$

878	S0016756813000162 doi: 10.1017/S0016756813000162
879	Dannberg, J., & Gassmöller, R. (2018). Chemical trends in ocean islands explained
880	by plume-slab interaction. Proceedings of the National Academy of Sciences.
881	115(17), 4351-4356. Retrieved from https://www.pnas.org/content/115/
882	17/4351 doi: 10.1073/pnas.1714125115
883	Dannberg, J., Gassmöller, R., Grove, R., & Heister, T. (2019, 05). A new formula-
884	tion for coupled magma/mantle dynamics. Geophysical Journal International.
885	219(1) 94–107 Retrieved from https://doi.org/10.1093/gii/ggz190 doi:
886	10 1093/gij/goz190
007	Dannherg J & Heister T (2016) Compressible magma/mantle dynamics: 3D
888	adaptive simulations in ASPECT Geophysical Journal International 207(3)
000	1343-1366 Betrieved from https://dx doi org/10 1093/gij/ggw329 doi:
800	10 1093/gij/ggw329
801	Dockman D. Pearson D. Heaman L. Gibson S. & Sarkar C. (2018) Tim-
891	ing and origin of magmatism in the sverdrup basin, northern canada—
892	implications for lithogeneric evolution in the high arctic large igneous
893	province (halin) Tectonombusics $7/2$ - $7/3$ 50–65 Retrieved from https://
094 905	www.sciencedirect.com/science/article/nii/S004019511830180X doi:
806	10 1016/i tecto 2018 05 010
090	Duvernay T Davies D R Mathews C R Cibson A H & Kramer S C
897	(2022) Continental magnetism: The surface manifestation of dynamic in-
898	teractions between cratonic lithosphere, mantle plumes and edge-driven con-
899	vection Geochemistry Geophysics Geosystems 23(7) e2022GC010363
900	Retrieved from https://agupubs_onlinelibrary_wiley_com/doi/
901	abs/10_1029/2022GC010363 (e2022GC010363 2022GC010363) doi:
902	10 1029/2022GC010363
903	Dessing A Gaina C & Brozena I M (2017) Building and breaking
904	a large igneous province: An example from the high arctic
905	ical Research Letters 1/(12) 6011-6019 Betrieved from https://
900	agunubs onlinelibrary wiley com/doi/abs/10 1002/2016GL072420 doi:
907	https://doi.org/10.1002/2016GL072420
900	Estrada S (2015) Geochemical and sr-nd isotope variations within cretaceous
909	continental flood-basalt suites of the canadian high arctic with a focus on
011	the bassel formation basalts of northeast ellesmere island International
912	Journal of Earth Sciences, 104, 1981-2005, doi: https://doi.org/10.1007/
913	s00531-014-1066-x
014	Estrada S. Damaske D. Henies-Kunst F. Schreckenberger B. Oakey G. N.
015	Pieniohn K Linnemann II (2016) Multistage cretaceous magmatism in
916	the northern coastal region of ellesmere island and its relation to the formation
917	of alpha ridge - evidence from aeromagnetic, geochemical and geochronological
918	data. Norwegian Journal of Geology, 96(2). doi: 10.17850/njg96-2-03
919	Estrada, S., & Henies-Kunst, F. (2004). Volcanism in the canadian high arctic
920	related to the opening of the arctic ocean. Zeitschrift der Deutschen Geologis-
921	chen Gesellschaft, 154, 579-603, doi: 10.1127/zdgg/154/2004/579
022	Estrada S Henies-Kunst F Melcher F & Tessensohn F (2010) Paleocene alka-
923	line volcanism in the pares strait region: evidence from volcanic pebbles. Inter-
924	national Journal of Earth Sciences, 99. doi: 10.1007/s00531-009-0432-6
925	Evenchick, C. A., Davis, W. J., Bédard, J. H. Hayward, N. & Friedman, R. M.
926	(2015, 09). Evidence for protracted High Arctic large igneous province mag-
927	matism in the central Sverdrup Basin from stratigraphy geochronology and
928	paleodepths of saucer-shaped sills. GSA Bulletin, 127(9-10), 1366-1390 Re-
929	trieved from https://doi.org/10.1130/B31190.1 doi: 10.1130/B31190.1
930	Galloway, J. M., Fensome, R. A., Swindles, G. T., Hadlari, T., Fath, J., Schröder-
931	Adams, C., Pugh, A. (2022). Exploring the role of high arctic large
	ignous province volcanism on early create course arctic forests Create course

933	Research, 129, 105022. Retrieved from https://www.sciencedirect.com/
934	science/article/pii/S0195667121002706 doi: https://doi.org/10.1016/
935	j.cretres.2021.105022
936	Hadlari, T., Dewing, K., Matthews, W. A., Alonso-Torres, D., & Midwinter, D.
937	(2018). Early triassic development of a foreland basin in the canadian high
938	arctic: Implications for a pangean rim of fire. <i>Tectonophysics</i> , 736, 75-84.
939	Retrieved from https://www.sciencedirect.com/science/article/pii/
940	S004019511830163X doi: https://doi.org/10.1016/j.tecto.2018.04.020
941	Heister, T., Dannberg, J., Gassmöller, R., & Bangerth, W. (2017). High accu-
942	racy mantle convection simulation through modern numerical methods. II:
943	Realistic models and problems. Geophysical Journal International, 210(2),
944	833-851. Retrieved from https://doi.org/10.1093/gji/ggx195 doi:
945	$10.1093/g_{1/ggx195}$
946	Heyn, B. H., & Conrad, C. P. (2022). On the relation between basal erosion of the
947	inthosphere and surface heat hux for continental plume tracks. Geophys-
948	active and a second definition of the second s
949	(2002)CL 08003 2022CL 08003) doi: 10 1020/2022CL 08003
950	Heyn B H Conrad C P & Trønnes B G (2020) How thermochemical piles
951	can (periodically) generate plumes at their edges Journal of Geophysical
953	Research: Solid Earth, 125(6), e2019JB018726. Retrieved from https://
954	agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019JB018726
955	(e2019JB018726 10.1029/2019JB018726) doi: 10.1029/2019JB018726
956	Ito, G. (2001). Reykjanes 'v'-shaped ridges originating from a pulsing and dehydrat-
957	ing mantle plume. Nature, 411(6838), 681–684. doi: 10.1038/35079561
958	Jackson, H. R., Dahl-Jensen, T., & working group, L. (2010, 07). Sedimentary and
959	crustal structure from the Ellesmere Island and Greenland continental shelves
960	onto the Lomonosov Ridge, Arctic Ocean. Geophysical Journal International,
961	182(1), 11-35.
962	Jakobsson, M., Mayer, L., Coakley, B., Dowdeswell, J. A., Forbes, S., Fridman, B.,
963	Weatherall, P. (2012). The international bathymetric chart of the arctic
964	ocean (1bcao) version 3.0. Geophysical Research Letters, 39(12). Retrieved
965	from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/
966	2012GL052219 doi: 10.1029/2012GL052219
967	Jokat, W., Ickrath, M., & O Connor, J. (2015). Seisinic transect across the
968	the amerasia basin arctic ocean <i>Combusical Research Letters</i> 10(10) 5047-
909	5051 Betrieved from https://agupubs.onlinelibrary.wiley.com/doi/abs/
971	10.1002/grl.50975 doi: https://doi.org/10.1002/grl.50975
972	Jones, S., Murton, B., Fitton, J., White, N., Maclennan, J., & Walters, R. (2014).
973	A joint geochemical–geophysical record of time-dependent mantle convec-
974	tion south of iceland. <i>Earth and Planetary Science Letters</i> , 386, 86–97.
975	Retrieved from http://www.sciencedirect.com/science/article/pii/
976	S0012821X13005402 doi: 10.1016/j.epsl.2013.09.029
977	Jones, S. M., White, N., & Maclennan, J. (2002). V-shaped ridges around
978	iceland: Implications for spatial and temporal patterns of mantle convec-
979	tion. Geochemistry, Geopysics, Geosystems, $3(10)$, 1059. Retrieved from
980	https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2002GC000361
981	doi: 10.1029/2002GC000361
982	Jowitt, S. M., Williamson, MC., & Ernst, R. E. (2014, 03). Geochemistry of the
983	130 to 80 Ma Canadian High Arctic Large Igneous Province (HALIP) Event
984	and Implications for Ni-Cu-PGE Prospectivity. Economic Geology, $109(2)$,
985	201-307. Retrieved from https://doi.org/10.2113/econgeo.109.2.281
986	uur. 10.2116/econgeo.109.2.201 Kontak D. I. Janson S. M. Dostal, I. Anabibald D. A. & Wessen T. V. (2001
987	Kontak, D. J., Jensen, S. M., Dostai, J., Archibaid, D. A., & Kyser, I. K. (2001,

988	08). CRETACEOUS MAFIC DYKE SWARM, PEARY LAND, NORTH-
989	ERNMOST GREENLAND: GEOCHRONOLOGY AND PETROLOGY. The
990	Canadian Mineralogist, 39(4), 997-1020. Retrieved from https://doi.org/
991	10.2113/gscanmin.39.4.997 doi: 10.2113/gscanmin.39.4.997
992	Kronbichler, M., Heister, T., & Bangerth, W. (2012). High accuracy mantle convec-
993	tion simulation through modern numerical methods. Geophysical Journal In-
994	ternational, 191, 12-29. Retrieved from http://dx.doi.org/10.1111/j.1365
995	-246X.2012.05609.x doi: 10.1111/j.1365-246X.2012.05609.x
996	Lawver, L. A., & Müller, R. D. (1994, 04). Iceland hotspot track. Geol-
997	ogy, 22(4), 311-314. Retrieved from https://doi.org/10.1130/0091
998	-7613(1994)022<0311:IHT>2.3.CD;2 doi: 10.1130/0091-7613(1994)022(0311:
999	IHT $2.3.CO;2$
1000	Lebedeva-Ivanova, N., Gaina, C., Minakov, A., & Kashubin, S. (2019). Arc-
1001	crust: Arctic crustal thickness from 3-d gravity inversion. <i>Geochemistry</i> ,
1002	Geophysics, Geosystems, 20(7), 3225-3247. Retrieved from https://
1003	agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2018GC008098 doi:
1004	https://doi.org/10.1029/2018GC008098
1005	Liu, J., Pearson, D. G., Wang, L. H., Mather, K. A., Kjarsgaard, B. A., Schaef-
1006	fer, A. J., Armstrong, J. P. (2021). Plume-driven recratonization of
1007	deep continental lithospheric mantle. Nature, 592(7856), 732–736. doi:
1008	10.1038/s41586-021-03395-5
1009	Maher, H. D., Jr. (2001). Manifestations of the cretaceous high arctic large igneous
1010	province in svalbard. The Journal of Geology, 109(1), 91-104. doi: 10.1086/
1011	317960
1012	Manión-Cabeza Córdoba, A., & Ballmer, M. D. (2021). The role of edge-driven
1013	convection in the generation of volcanism – part 1: A 2d systematic study.
1014	Solid Earth. 12(3), 613-632. Retrieved from https://se.copernicus.org/
1015	articles/12/613/2021/ doi: 10.5194/se-12-613-2021
1016	Manión-Cabeza Córdoba, A., & Ballmer, M. D. (2022). The role of edge-driven
1017	convection in the generation of volcanism – part 2: Interaction with mantle
1018	plumes, applied to the canary islands. Solid Earth, 13(10), 1585–1605. Re-
1019	trieved from https://se.copernicus.org/articles/13/1585/2022/ doi:
1020	10.5194/se-13-1585-2022
1021	Martos, Y. M., Jordan, T. A., Catalán, M., Jordan, T. M., Bamber, J. L., &
1022	Vaughan, D. G. (2018). Geothermal heat flux reveals the iceland hotspot
1023	track underneath greenland. Geophysical Research Letters, 45(16), 8214–8222.
1024	Retrieved from https://agupubs.onlinelibrary.wiley.com/doi/abs/
1025	10.1029/2018GL078289 doi: 10.1029/2018GL078289
1026	Midtkandal, I., Faleide, J. I., Faleide, T. S., Serck, C. S., Planke, S., Corseri, R.,
1027	Nystuen, J. P. (2020). Lower cretaceous barents sea strata: epicontinental
1028	basin configuration, timing, correlation and depositional dynamics. <i>Geological</i>
1029	Magazine, 157(3), 458–476. doi: 10.1017/S0016756819000918
1030	Minakov, A., Yarushina, V., Faleide, J. I., Krupnova, N., Sakoulina, T., Dergunov,
1031	N., & Glebovsky, V. (2018). Dyke emplacement and crustal structure
1032	within a continental large igneous province, northern barents sea. Geo-
1033	logical Society, London, Special Publications, 460(1), 371–395. Retrieved
1034	from https://www.lyellcollection.org/doi/abs/10.1144/SP460.4 doi:
1035	10.1144/SP460.4
1036	Naber, T., Grasby, S., Cuthbertson, J., Rayner, N., & Tegner, C. (2020, 12). New
1037	constraints on the age, geochemistry, and environmental impact of High Arctic
1038	Large Igneous Province magmatism: Tracing the extension of the Alpha Ridge
1039	onto Ellesmere Island, Canada. GSA Bulletin, 133(7-8), 1695–1711. Retrieved
1040	from https://doi.org/10.1130/B35792.1 doi: 10.1130/B35792.1
1041	Negredo, A. M., van Hunen, J., Rodríguez-González, J., & Fullea, J. (2022). On
1042	the origin of the canary islands: Insights from mantle convection modelling.
	· · · · · · · · · · · · · · · · · · ·

1043	Earth and Planetary Science Letters, 584, 117506. Retrieved from https://
1044	www.sciencedirect.com/science/article/pii/S0012821X2200142X doi:
1045	10.1016/j.epsl.2022.117506
1046	Oakey, G., & Saltus, R. (2016). Geophysical analysis of the alpha–mendeleev ridge
1047	complex: Characterization of the high arctic large igneous province. Tectono-
1048	physics, 691, 65-84. Retrieved from https://www.sciencedirect.com/
1049	science/article/pii/S0040195116303304 (SI:Circum-Arctic Evolution)
1050	doi: https://doi.org/10.1016/j.tecto.2016.08.005
1051	Parnell-Turner, R., White, N., Henstock, T., Murton, B., Maclennan, J., & Jones,
1052	S. M. (2014). A continuous 55-million-year record of transient mantle
1053	plume activity beneath iceland. Nature Geoscience, 7(12), 914–919. doi: 10.1028/mmc.2281
1054	10.1050/lige02201 Desse V. Dresher S. Stephenson D. & Zhang V. (2014) Aretic lithe
1055	sphere a review Testen on husing 628 1.25 Retrieved from https://
1056	www.sciencedirect.com/science/article/nii/S0040195114003060 doi:
1057	https://doi.org/10.1016/j.tecto.2014.05.033
1050	Polteau S Hendriks B W Planke S Ganerød M Corfu F Faleide I I
1059	Myklebust B (2016) The early cretaceous harents sea sill complex:
1061	Distribution. 40ar/39ar geochronology, and implications for carbon gas for-
1062	mation. Palaeogeography. Palaeoclimatology. Palaeoecology, 441, 83–95.
1063	Retrieved from https://www.sciencedirect.com/science/article/pii/
1064	S0031018215003673 (Impact, Volcanism, Global changes and Mass Extinc-
1065	tions) doi: $10.1016/j.palaeo.2015.07.007$
1066	Saumur, B., Dewing, K., & Williamson, MC. (2016). Architecture of the canadian
1067	portion of the high arctic large igneous province and implications for magmatic
1068	ni–cu potential. Canadian Journal of Earth Sciences, $53(5)$, $528-542$. doi:
1069	10.1139/cjes-2015-0220
1070	Schaeffer, A. J., & Lebedev, S. (2013, 04). Global shear speed structure of the upper
1071	mantle and transition zone. Geophysical Journal International, 194(1), 417-
1072	449. Retrieved from https://doi.org/10.1093/gji/ggt095 doi: 10.1093/
1073	$g_{\rm J1}/ggt095$
1074	tiam in control gritaborgon Cocchamistry Coordinate According 22(11)
1075	o2021CC010300 Betrieved from https://agupubs.onlinelibrary.uilev
1076	$com/doi/abs/10_1029/2021GC010300_(e2021GC010300_2021GC010300)$ doi:
1077	10 1029/2021GC010300
1079	Shephard, G. E., Müller, B. D., & Seton, M. (2013, September). The tectonic evo-
1080	lution of the Arctic since Pangea breakup: Integrating constraints from surface
1081	geology and geophysics with mantle structure. <i>Earth-Science Reviews</i> , 124,
1082	148-183. Retrieved 2022-10-17, from https://linkinghub.elsevier.com/
1083	retrieve/pii/S0012825213001104 doi: 10.1016/j.earscirev.2013.05.012
1084	Shephard, G. E., Trønnes, R. G., Spakman, W., Panet, I., & Gaina, C. (2016).
1085	Evidence for slab material under greenland and links to cretaceous high arc-
1086	tic magmatism. Geophysical Research Letters, $43(8)$, $3717-3726$. Retrieved
1087	from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1002/
1088	2016GL068424 doi: 10.1002/2016GL068424
1089	Steffen, R., Audet, P., & Lund, B. (2018). Weakened lithosphere beneath green-
1090	land inferred from effective elastic thickness: A hot spot effect? Geo-
1091	physical Research Letters, 45(10), 4/35-4/42. Retrieved from https://
1092	agupubs.onlinelibrary.Wiley.com/dol/abs/10.1029/201/GL0/6885 dol: https://doi.org/10.1020/2017CL076885
1004	Steinherger B. Bredow F. Labedev S. Schoeffer A. & Torsvik T. H. (2010)
1094	Widespread volcanism in the greenland-north atlantic region explained
1095	by the iceland plume, Nature Geoscience, 12(1), 61–68. doi: 10.1038/
1097	s41561-018-0251-0;10.1038/s41561-018-0251-0

1098	Sun, P., Niu, Y., Guo, P., Duan, M., Wang, X., Gong, H., & Xiao, Y. (2020).
1099	The lithospheric thickness control on the compositional variation of conti-
1100	nental intraplate basalts: A demonstration using the cenozoic basalts and
1101	clinopyroxene megacrysts from eastern china. Journal of Geophysical Re-
1102	search: Solid Earth, 125(3), e2019JB019315. Retrieved from https://
1103	agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019JB019315
1104	$(e2019JB019315\ 2019JB019315)$ doi: $10.1029/2019JB019315$
1105	Tarduno, J. (1998). The high arctic large igneous province. <i>Third International</i>
1106	Conference on Arctic margins [ICAM III], Celle, Germany, Fed. Inst. Geosci.
1107	Nat Res. Abstract
1108	Tegner, C., Storey, M., Holm, P., Thorarinsson, S., Zhao, X., Lo, CH., & Knud-
1109	sen, M. (2011). Magmatism and eurekan deformation in the high arctic
1110	large igneous province: 40ar–39ar age of kap washington group volcanics,
1111	north greenland. Earth and Planetary Science Letters, 303(3), 203–214.
1112	Retrieved from https://www.sciencedirect.com/science/article/pii/
1113	S0012821X10008204 doi: 10.1016/j.epsl.2010.12.047
1114	Thorarinsson, S. B., Holm, P. M., Tappe, S., Heaman, L. M., & Tegner, C. (2011).
1115	Late cretaceous–palaeocene continental rifting in the high arctic: U–pb
1116	geochronology of the kap washington group volcanic sequence, north green-
1117	land. Journal of the Geological Society, 168(5), 1093-1106. Retrieved from
1118	https://www.lyellcollection.org/doi/abs/10.1144/0016-76492011-018
1119	doi: 10.1144/0016-76492011-018
1120	Trettin, H. E. (n.d.). Geology of canada, no. 3; geology of the innuitian orogen
1121	and arctic platform of canada and greenland. (569 pp.). Geological Survey of
1122	Canada(3).
1123	Villeneuve, M., & Williamson, MC. (2006). 40Ar-39Ar dating of matic magnatism
1124	from the Sverdrup Basin Magmatic Province. Scott, R.A., Thurston, D.K.
1125	(Eds.), ICAM IV Proceedings, Dartmouth, Nova Scotia, 206-215.